Contents lists available at ScienceDirect

Applied Energy

journal homepage: www.elsevier.com/locate/apenergy

Big data mining for the estimation of hourly rooftop photovoltaic potential and its uncertainty



^a Solar Energy and Building Physics Laboratory, Ecole Polytechnique Fédérale de Lausanne, Switzerland
^b Environmental Change Institute, University of Oxford, Oxford, United Kingdom

HIGHLIGHTS

- Hourly profiles of photovoltaic potential are estimated for 9.6 M Swiss rooftops.
- The data mining approach combines Machine Learning and Geographic Information Systems.
- Uncertainties are quantified and propagated throughout all stages of the estimation.
- $\bullet\,$ Switzerland's annual rooftop PV potential is estimated at 24 $\pm\,$ 9 TWh.
- It may cover up to 43% of the national electricity demand (in 2018).

ARTICLE INFO

Keywords: Rooftop photovoltaic potential Spatio-temporal modelling Big data mining Uncertainty estimation Machine Learning

ABSTRACT

The large-scale deployment of photovoltaics (PV) on building rooftops can play a significant role in the transition to a low-carbon energy system. To date, the lack of high-resolution building and environmental data and the large uncertainties related to existing processing methods impede the accurate estimation of large-scale rooftop PV potentials. To address this gap, we developed a methodology that combines Machine Learning algorithms, Geographic Information Systems and physical models to estimate the technical PV potential for individual roof surfaces at hourly temporal resolution. We further estimate the uncertainties related to each step of the potential assessment and combine them to quantify the uncertainty on the final PV potential. The methodology is applied to 9.6 million rooftops in Switzerland and can be transferred to any large region or country with sufficient available data. Our results suggest that 55% of the total Swiss roof surface is available for the installation of PV panels, yielding an annual technical rooftop PV potential of 24 ± 9 TWh. This could meet more than 40% of Switzerland's current annual electricity demand. The presented method for an hourly rooftop PV potential and uncertainty estimation can be applied to the large-scale assessment of future energy systems with decentralised electricity grids. The results can be used to propose effective policies for the integration of rooftop photovoltaics in the built environment.

1. Introduction

The decarbonisation of the energy system plays an important role in fulfilling the ambitious emission targets set by the Paris Agreement [1]. In this context, the large-scale deployment of rooftop-mounted photovoltaics (RPV) has attracted increasing attention in recent years [2]. A quantitative assessment of the potential electricity generation from RPV is essential to formulate effective incentive policies for their integration in the built environment. This requires accurate input data at a high spatial and temporal resolution in order to characterize regional differences and to assess the seasonal and intra-day variation of the potential generation [3]. In addition, the analysis of RPV potential involves several uncertainties, which need to be quantified to facilitate interpretation of the results for policy making.

Currently, there is no methodology that estimates the large-scale RPV potential at a high spatio-temporal resolution and also addresses the systematic propagation of uncertainties arising from the modelling process. This paper contributes to fill this gap by adapting state of the art methods for the assessment of RPV potential in order to quantify and combine different sources of uncertainty. We further use Machine Learning to incorporate information that is only available in parts of the study region of Switzerland.

* Corresponding author.

E-mail address: alina.walch@epfl.ch (A. Walch).

https://doi.org/10.1016/j.apenergy.2019.114404

Received 13 August 2019; Received in revised form 22 November 2019; Accepted 14 December 2019

0306-2619/ @ 2020 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/BY-NC-ND/4.0/).







Constrained by the availability of building and environmental data at a high spatial resolution, most existing studies on RPV potential are carried out at district or city scale only, including case studies in the US [4], Canada [5], Portugal [6] or Germany [7]. Few studies analyse an entire region or country, for example in the US [8,9], Spain [10], or Saudia Arabia [11]. Many PV assessment studies use monthly or yearly solar radiation data in order to derive a large-scale PV potential [12]. This data is used to quantify the available area to install PV [13] and to discuss the economic feasibility of PV scenarios [14]. While these granularities provide a relatively accurate estimation of the annual RPV potential, an hourly or higher temporal resolution is needed to assess the intra-day variation of the generation. Such temporal resolutions are used at the scale of cities or municipalities [15,16] and serve to validate the estimations against measurements [17], to compare PV technologies [18], or to simulate energy systems with high shares of PV [19]. However, an hourly resolution is used very rarely at regional or national scale, in which case the available area to install PV is not addressed [20].

Reasons for the lack of national-scale studies at an hourly temporal resolution are the computational challenges associated with the processing of the required input datasets as well as the handling of missing data and data that is not available in the entire study region. To address these challenges, state of the art data processing techniques for estimating RPV potentials include physical models, geographic information systems (GIS), image processing and Machine Learning (ML). Physical models are used to compute the solar radiation on tilted surfaces, as well as the module and inverter efficiencies [21]. GIS is applied to estimate shading effects and the sky view factor [22,23]. The available area for installing PV is estimated in some studies using image recognition [24,25], while other methods use ML [26,27]. The recent computational and methodological advances enable the integration of all mentioned aspects in high-resolution PV assessments at the national scale. Missing data, in particular for predicting the available area for installing PV, is typically handled using constant coefficients and expert knowledge [28] or sampling techniques [29]. Data with partial spatial coverage, i.e. information that is available only in parts of the study area, is successfully used in [26,27] by applying ML. This method can be further improved by using different urban features and a larger training dataset.

To date, little research has quantified uncertainties for RPV potential assessments. Some studies address the uncertainties related to photovoltaic yield predictions for individual case studies [30,31]. In large-scale studies, confidence intervals are used in [20] to quantify the uncertainty related to the solar radiation, while they are used in [27] to assess the available area for installing PV. The combination of different sources of uncertainty has been addressed through a scenario-based analysis for a local case study [32] or by means of a sensitivity analysis [33]. Izquierdo et al. [34] provide a statistical propagation of uncertainty, which however focuses only on the available area for PV installation. To the best of our knowledge, currently no methodology exists which quantifies and combines different sources of uncertainty related to the solar radiation and the available area that yields an uncertainty estimate on the technical RPV potential.

Our big data mining approach contributes to the existing literature by providing a methodology for a large-scale RPV potential and

uncertainty estimation in hourly temporal resolution and a spatial resolution of individual roof surfaces. For this purpose, we combine state of the art physical models and GIS processing techniques with ML in order to quantify (i) the spatio-temporal variation of the horizontal radiation, (ii) the effects of surrounding trees and buildings on roof shading and the sky view factor, (iii) the impact of roof geometry and roof superstructures, for example dormers and chimneys, on the available area for installing PV panels, and (iv) the temperature dependence of the PV module efficiency. We further present a systematic quantification of the uncertainties by treating each variable involved in the modelling process as a randomly distributed variable. This allows to combine the uncertainties using their statistical distribution in order to obtain a total uncertainty on the technical potential estimate. The application of our method to 3.7 million buildings in Switzerland results in the first national-scale dataset of PV potential and uncertainty in hourly temporal resolution for individual rooftops.

The paper is structured as follows. Section 2 introduces the datasets used in the study. Section 3 describes the methodology for the computation of the RPV potential and the uncertainty propagation. Section 4 presents the results of the individual processing steps as well as the final technical RPV potential. A sensitivity analysis is performed to identify the most impactful steps of the estimation process. Section 5 discusses the methodological and practical contribution of the work and outlines its limitations and applications. Section 6 presents the conclusions and gives an outlook to future applications of the developed method.

2. Data

Using big data mining techniques for the estimation of large-scale RPV potential requires the availability of accurate and high-resolution environmental and building datasets. Our approach combines large sets of meteorological data, building data and digital surface models at different resolutions and spatial coverage. Switzerland has been selected as case study area due to its high data availability.

2.1. Meteorological data

We use four types of meteorological data, namely hourly global, direct solar horizontal radiation (in W/m²), daily surface reflectance (albedo), and daily maximum temperature (in °C), recorded during 12 years from 2004 to 2015. The specifications of all datasets are summarized in Table 1. The solar radiation and albedo data is derived from Meteosat Second Generation (MSG) satellite observations using the Heliomont algorithm [35]. Satellite data is preferred over data from measurement stations as it provides a better spatial coverage with an increased resolution, it has a very low missing data ratio (<1%) and it shows a negligible bias [36]. The daily maximum temperature data is the result of a gridded interpolation of near-surface air temperature measurements with errors below 1°C at urban altitudes [37]. As the PV panel efficiency decreases with temperature, using the daily maximum temperature corresponds to the least optimal PV performance.

We average the 12 years of data to obtain an annual mean dataset, i.e. 8760 time steps for solar radiation and 365 time steps for albedo and temperature. This reduces the variability of the meteorological data

Table 1

Meteorological data used in the study.

Data	Spatial res.	Time	Range	Source
Global horizontal radiation	1.25 deg. min. ^a	hourly	2004–2015	MeteoSwiss [38]
Direct horizontal radiation	1.25 deg. min. ^a	hourly	2004–2015	MeteoSwiss [38]
Surface albedo	1.25 deg. min. ^a	daily	2004–2015	MeteoSwiss [38]
Maximum temperature	1 km ²	daily	2004–2015	MeteoSwiss [39]

^a Deg. min. denotes degree minutes on a longitude-latitude grid. 1.25 deg. min. corresponds to approximately (1.6×2.3) km².

 Table 2

 Building data used in the study.

0				
Data	Coverage	Spatial res.	Creation	Source
Roof surfaces Register of	Switzerland Switzerland	Rooftops Buildings	2010–2016 2015	Sonnendach [22] SwissStat [40]
Superstructures	Geneva Canton	Rooftops	2005-2011	SITG [41]

and allows the estimation of long-term PV potential without bias due to extreme meteorological events of a specific year.

2.2. Building data

The computation of the RPV potential is based on a national dataset of building roofs. It contains around 9.6 million vector polygons representing all roofs in Switzerland's 3D building cadastre (LOD 2) with the roof tilt, aspect (both in degree) and area (in m²). This dataset is combined with the national register of buildings and dwellings (RBD), which gives information on the building's footprint, construction period, number of floors and building type. To account for obstructing objects on rooftops (superstructures), which impede the installation of PV panels, we use a dataset derived from detailed city GML data (LOD 4) available only in the Canton of Geneva. It contains vector polygons that represent superstructures such as dormers and chimneys and covers nearly 38,000 roofs. Table 2 summarizes the building data.

2.3. Digital elevation models

Besides meteorological and building data, we use a Digital Terrain Model (DTM) and Digital Surface Models (DSM) derived from Light Detection and Ranging (LiDAR) data, which are summarized in Table 3. From the DTM, terrain characteristics including altitude, slope and curvature have been computed by Robert et al. [42]. The DSM is used to account for the effects of landscape characteristics on the PV potential, which requires the most up-to-date and accurate LiDAR data. We hence use a DSM at national scale in (2×2) m² resolution, denoted as DSM_{2m}, as well as a higher-resolution DSM of (0.5×0.5) m² available for the Canton of Geneva only, denoted as DSM_{50cm}.

3. Methodology

To estimate the RPV potential and its uncertainty, we propose a methodology that combines Machine Learning, GIS processing and physical models for the treatment of large spatio-temporal datasets, such as those presented in Section 2. Uncertainties are quantified from the statistical distribution of the variables involved in the potential estimation in the form of standard deviations. They are assessed for various modelling steps and combined in order to obtain an uncertainty on the PV potential.

The method, illustrated in Fig. 1, is based entirely on open-source software and adopts an hierarchical approach used in several related studies [21,26,34]. Its steps include (i) the *physical potential*, driven by the horizontal solar radiation, (ii) the *geographic potential*, accounting for the impact of the built environment, and (iii) the *technical potential*, defined as the potential electricity generation. The variables involved in each stage of the model and their relationships are explained below.

Table 3

Digital elevation models used in th	e study.
-------------------------------------	----------

Data	Coverage	Spatial resolution	Creation	Source
DTM	Switzerland	$(2 \times 2) m^2$	2010–2016	SwissTopo [43]
DSM _{2m}	Switzerland	$(2 \times 2) m^2$	2000–2008	SwissTopo [44]
DSM _{50cm}	Geneva Canton	$(0.5 \times 0.5) m^2$	2013	SITG [45]

The following subsections will detail the methodological contributions for the individual processing steps and the quantification of uncertainty.

The physical potential is defined as the horizontal solar radiation at the earth's surface (G_h) for each time step *t*. The G_h is composed of a direct beam component (G_B) and a diffuse component (G_D) such that [46]:

$$G_h(t) = G_B(t) + G_D(t) \tag{1}$$

(

The horizontal radiation is computed at a monthly-mean hourly (MMH) temporal resolution and a spatial resolution of (200×200) m², which is chosen as a trade-off between topographic detail and computational complexity as suggested in [27,34]. Each MMH time step represents an average value at a given hour of each month, across all days of the month. This leads to 288 distinct time steps, i.e. 24 h for each of the 12 months. Using MMH values instead of 8760 hourly time steps allows to reduce the computational cost by a factor of 30, while preserving the daily and seasonal patterns of the average PV potential. Any deviation from the MMH values is however covered by their uncertainty.

The geographic potential accounts for the rooftop geometry, for superstructures, for shading effects and for the sky visibility. It is estimated for each roof surface considering the tilted radiation (G_t) and the available roof area for PV panel installation (A_{PV}) [27], such that:

$$G_t(t) = (1 - S_{sh}(t)) * G_{Bt}(t) + SVF * G_{Dt}(t) + G_{Rt}(t)$$
(2)

$$A_{PV} = A_t * C_{pv} * (1 - C_{sh})$$
(3)

where G_{Bt} , G_{Dt} and G_{Rt} are direct, diffuse and reflected tilted radiation components, *SVF* is the sky view factor, A_t is the tilted roof area, C_{pv} is referred to as the *panelled area coefficient* and C_{sh} and S_{sh} are referred to as the *shaded area coefficient* and the *hourly shading fraction* of the rooftop, respectively. The C_{sh} represents the fraction of roof surface that is unshaded in less than 40% of the daylight hours and is hence unsuitable for PV installation (see Section 3.5). The $S_{sh}(t)$ denotes the portion of the remaining roof area $(1 - C_{sh})$ that is shaded at each time step *t*. We treat C_{sh} and C_{pv} , the proportion of roof area available to install PV panels, as independent factors. They are separately computed and it is assumed that no relevant overlap exists between the two.

The technical potential (E_{PV}) is the electricity output of each roof surface. It is obtained from the geographic potential, the panel efficiency (η_{PV}) and the performance factor (*PF*), which accounts for inverter efficiency and other losses such as soiling or degradation, such that [27]:

$$E_{PV} = G_t(t) * A_{PV} * \eta_{PV}(t) * PF(t)$$

$$\tag{4}$$

Table 4 summarizes the variables introduced above and the method used to model each of them.

3.1. Physical models

State of the art hourly physical models are used to (i) obtain the tilted radiation components (G_{Bt} , G_{Dt} , G_{Rt}) and (ii) to quantify the module and inverter efficiency. Both models, shown as orange boxes in Fig. 1, are summarized in Appendix A. We use the anisotropic Perez model [47] to estimate G_{Dt} , which is the overall most accurate diffuse radiation model [46,48]. The G_{Rt} is computed using monthly mean albedo data (see Section 3.3). This allows to account for snow cover in high-altitude locations, where a large albedo significantly increases the PV production on steep roofs [49]. The efficiencies are calculated for each time step *t* from the tilted radiation and the ambient temperature using the *PVWatts* model [50]. We use daily maximum temperature (see Table 1) to obtain a conservative estimate of the panel efficiency.



Fig. 1. Workflow for modelling of hourly solar PV potential.

Summary of the parameters used in the estimation of technical RPV potential. The dimension refers to *P* for pixels of (200×200) m², *R* for roof surfaces and *t* for time steps. PM denotes the use of physical models.

	$G_{h,B,D}$	$G_{Bt,Dt,Rt}$	S_{sh}	SVF	C_{sh}	C_{pv}	η_{PV}, PF
Dimension	$P \times t$	$R \times t$ W/m ² PM 3.1	R × t	R	R	R	R × t
Unit	W/m ²		-		-	-	-
Method	ML		GIS	GIS	GIS + ML	GIS + ML	PM
Section	3.3		3.5	3.6	3.5	3.4	3.1

3.2. Machine Learning

Machine Learning is used here (i) to obtain the horizontal radiation and the albedo at the (200×200) m² output grid from lower-resolution satellite data, (ii) to estimate C_{sh} and (iii) to account for the roof area covered by superstructures in the estimation of C_{pv} . We use supervised regression algorithms, which learn from a training set that contains pairs of inputs (features) and their known output values (targets). The models are then applied to predict unknown outputs for a new set of features.

We consider two ML algorithms, Random Forests (RF) [51] and Extreme Learning Machine Ensembles (ELM-E) [52,53]. Both are ensemble algorithms which predict a target variable by averaging the results of multiple estimators (decision trees/ELMs) that have been trained on random re-samples of the training data [54]. Ensembles are well-suited for the estimation of uncertainties, which give a useful indication of the model's accuracy [55]. The ELM-E is very efficient at

modelling large datasets [56], so we use this algorithm to estimate the hourly solar radiation and the monthly albedo. Preliminary work has shown that the RF outperforms the ELM-E for higher-dimensional datasets with less training data. The RF is hence used to estimate C_{sh} and C_{pv} .

To optimize the performance of each ML model, it is necessary to firstly select its features, and secondly to choose the parameters defining the structure of each model through hyper-parameter tuning. In the following sections, we will focus on the feature selection and the performance of the optimized ML models. Information regarding the tuning of the algorithms is provided in Appendix B.

3.3. Horizontal solar radiation and albedo

We propose an ML-based approach to estimate the horizontal radiation and the surface albedo for pixel maps of (200×200) m² resolution, as presented in the blue box in Fig. 1-A. The ML models yield the global (G_h) and direct (G_B) horizontal radiation for each MMH time step *t* and the albedo (ρ) as monthly mean values. The diffuse radiation (G_D) is obtained from the estimated G_h and G_B by applying Eq. (1). The use of ML allows to model the complex spatial patterns while being more efficient than other spatial interpolation techniques such as kriging [57].

The targets for the ML models of G_h and G_B are the satellite-derived global and direct radiation, which are split into 12 monthly subsets. The initially considered set of features are the mean hour of each month, the geographic coordinates (x, y) as well as several terrain features including altitude (z), terrain slope and curvature [42]. A DTM is used to derive these terrain features for the satellite data and for the



Fig. 2. (a) Feature importance per hour for G_h for June and (b) December for six terrain features, namely altitude (z), longitude (x), latitude (y), medium-scale curvature (medDoG), small-scale curvature (smallDoG), terrain slope in east-west (big_EW) and north-south (big_NS) direction.

Test mean-squared error (MSE) for the estimation of G_h and G_B .

MSE	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
G_h	0.10	0.07	0.07	0.06	0.06	0.04	0.06	0.05	0.12	0.10	0.10	0.06
G_B	0.17	0.12	0.18	0.17	0.21	0.14	0.18	0.12	0.26	0.16	0.21	0.11

 (200×200) m² output grid. To better model the spatio-temporal patterns of G_h and G_B , we analyze the spatial features for each time step using the RF feature importance score [58]. Fig. 2 shows their pattern for the months of June and December. The feature importance exhibits a strong intra-day variation for x, y and z, which may be explained by the impact of the Swiss alpine terrain. All other features have an overall low importance throughout the day. Including these in the model does not improve its performance further. Four features are hence selected for the final model: hour, x, y, z. Fig. 2 refers to G_h , but G_B shows a similar trend.

Separate ELM-E are trained for each month for G_h and G_B . Their tuning procedure and the resulting hyper-parameter values of the models are provided in Appendix B. Table 5 shows the mean-squared error (MSE) between the targets and predictions for a random 20% of the satellite coordinates (test set), which are excluded from training. Each entry represents one trained ELM-E. The model performance is overall better for G_h than for G_B , as it is harder to model G_B from the given set of features. To estimate ρ from the satellite-derived daily albedo measurements, we train one ELM-E with similar features as described above, namely x, y, z and month. The model is tuned in a similar fashion as that of G_h and G_B and yields a test MSE of 0.10, which lies in the range of the values obtained for G_h .

3.4. Available area for PV panel installation

The *panelled area coefficient* C_{pv} describes the available roof area for PV installation considering the roof geometry and superstructures. To compute C_{pv} , we use a geospatial algorithm in combination with ML, as represented by the green and blue box yielding C_{pv} in Fig. 1-B. The geospatial algorithm, detailed in Appendix C, virtually installs PV panels by projecting rectangular polygons onto the tilted roofs as shown in Fig. 3. The C_{pv} is then obtained from the number of installed panels. The panels are installed in both horizontal and vertical alignments, as no alignment has technical advantages over the other and both are wide-spread. The configuration with a higher number of panels is selected for each roof.

The data on roof superstructures is however only available in one of Switzerland's 26 cantons (Geneva). We hence train an ML model to



Fig. 3. Output of the virtual installation of PV panels after removing roof superstructures. The best configuration of vertically and horizontally oriented panels is selected. Panels on flat roofs are placed in south-facing rows and tilted at an optimal angle of 30°.

Table 6

Features considered in the ML models to estimate C_{pv} and C_{sh} . ¹The building density is computed as the number of building coordinates within a 100 m radius of each roof. ²The roof perimeter, shape index (perimeter per area) and vertex count are derived from the polygon data.

Computed features	Roof features	Building features
C_{pv}^F / C_{sh}^F Panel tilt (not used for C_{sh}) Build. density ¹	Tilted area Tilt angle Aspect angle Perimeter ² Shape index ² Vertex count ²	Footprint area Building type Construction period Number of floors

estimate the change in C_{pv} due to the area covered by superstructures, which is applied to the rooftops of the remaining 25 cantons. In our case, the training of the ML model is performed using the rooftop dataset with superstructure information in the Canton of Geneva (see Section 2). The fraction of their area on which virtual panels are installed provides the target for the ML algorithm (C_{pv}^T) . In addition, we define C_{pv}^F as the fraction of the area on which virtual panels may be installed if no superstructures are removed from the roof polygons. This C_{pv}^F is one feature of the ML model. The full set of the considered features is listed in Table 6. Fig. 4 shows the feature importance for all inputs. While the top six features have the highest importance, the complete set of features is used for modelling C_{pv} as this has been found to improve the estimation precision.

Table 7 reports the residuals between C_{pv}^{F} and C_{pv}^{T} (baseline), as well as the cross-validation error between the estimated C_{pv} and the target C_{pv}^{T} in the case of additive bias correction (MBE +) and for using the tuned RF model (see Appendix B for details). The bias correction is performed by adding the mean bias error (MBE) between C_{pv}^{F} and C_{pv}^{T} to all samples. We compare the root mean squared error (RMSE), mean absolute error (MAE), MBE and the R² coefficient of determination. All methods remove the negative baseline bias of 17%. The RF outperforms the bias correction, as it achieves lower errors and a higher R². It is hence used to estimate the C_{pv} used in Eq. (3).

3.5. Shading effects

To quantify shading effects from surrounding buildings and trees, we use a shadow casting approach [20–23]. In contrast to the existing studies, we account for the shading effects in two ways: First, strong shading effects may render parts of a roof unsuitable for installing PV.



Fig. 4. Feature importance of all features considered for the estimation of C_{pv} .

Errors for estimating C_{pv} . We compare the residuals between C_{pv}^{P} and C_{pv}^{T} (baseline) with the cross-validation error between the estimated C_{pv} and the target C_{pv}^{T} in the case of bias correction (MBE+) and for using the Random Forest model (RF).

	RMSE	MAE	MBE	\mathbb{R}^2
Baseline	0.23	0.17	-0.17	-0.10
MBE +	0.15	0.12	0.00	0.52
RF	0.12	0.09	0.00	0.69

Second, some of the suitable area may be shaded at particular hours and reduce the PV potential for these hours. We hence propose to distinguish between the *shaded area coefficient* (C_{sh}), the fraction of roof area which is unsuitable due to shading, and the *hourly shading fraction* ($S_{sh}(t)$), which is computed for each time step as the shaded portion of the remaining roof.

The shadow casting approach (green box in Fig. 1-B) models shadows on building roofs for each pixel of a DSM at a given sun position. It is used to produce hourly shadow maps for the representative day of each month (close to day 15 [23]) and hence yields results at the same temporal resolution as the solar radiation. Fig. 5a shows an example of these maps. They are binary maps, with a 0 (dark areas) representing a cast shadow and a 1 (yellow areas) indicating direct sun exposure. The shadow maps are averaged for all daylight hours, resulting in the mean illumination map shown in Fig. 5b. Its values represent the percentage of time steps for which each pixel is unshaded.

All areas with a mean illumination lower than the average value across all roofs (red areas) are considered unsuitable. We compute this average value as 42.7% and hence use a rounded threshold of 40% mean illumination to exclude unsuitable areas. Dividing the red areas in Fig. 5b by the total roof area yields C_{sh} , while S_{sh} is the fraction of zeros in Fig. 5a on the suitable (non-red) surfaces at each time *t*. The GIS algorithm for the shading effects is given in Appendix C.

Two datasets are available to compute C_{sh} and S_{sh} : the national DSM_{2m} and the DSM_{50cm} for the canton of Geneva (see Table 3). Our method aims at enhancing the national-scale C_{sh} and S_{sh} from the DSM_{2m} based on knowledge extracted from the DSM_{50cm} , which is more recent and detailed. For this task, we denote the values extracted from the DSM_{2m} as C_{sh}^{F} , S_{sh}^{F} (features) and those obtained from the DSM_{50cm} as C_{sh}^{T} , S_{sh}^{T} (targets). Their comparison yields an MBE of 2.7% between S_{sh}^{F} and S_{sh}^{T} and of 8.9% between C_{sh}^{F} and C_{sh}^{T} . As the MBE for S_{sh} is small compared to the order of magnitude of other uncertainties, we use an additive bias correction (MBE +) to correct S_{sh}^{F} at the national scale, which is performed separately for each time step.

The C_{sh} shows a larger bias, so C_{sh}^F systematically underestimates the shaded area coefficient. This leads to an overestimation of available area. As a mean bias correction (MBE+) increases the MAE (see Table 8), we apply an ML model (RF) to estimate C_{sh} , illustrated by the blue box yielding C_{sh} in Fig. 1-B. The features that are used to estimate

Table 8

Errors for estimating C_{sh} . Baseline is the error between C_{sh}^F and C_{sh}^T . The considered models are MBE + and RF.

	RMSE	MAE	MBE	\mathbb{R}^2
Baseline	0.23	0.13	0.09	0.12
MBE +	0.21	0.14	0.00	0.25
RF	0.18	0.12	0.00	0.44



Fig. 6. Feature importance of all features considered for the estimation of C_{sh} .

 C_{sh} are listed in Table 6. Fig. 6 shows the feature importance for C_{sh} . Interestingly, the building type has a high importance. This may be due to the different objects that cause shading on different types of buildings.

The errors for estimating C_{sh} are shown in Table 8. While the improvement of the MAE when using ML is relatively small, we do observe a large improvement in the R² coefficient. Unpredictable factors, such as discrepancies between the DSM_{2m} and the DSM_{50cm} or mismatches between the roof polygons and the DSMs, keep even the improved R² coefficient rather low. These errors will be reflected in the results by an increased uncertainty.

3.6. Sky view factor

The sky view factor (SVF), representing the visible proportion of the sky, is computed by combining the vertical elevation angles of the horizon for a discretized set of directions [59]. We find 32 equally spaced directions to be sufficient for a precise estimation of the SVF, which we have implemented according to Appendix C. Using the DSM_{2m} and the DSM_{50cm}, we compute a national SVF^F and a "true" SVF^T in Geneva, respectively. The MBE between the two variables is 7.16%. This error of SVF^F with respect to SVF^T follows no identifiable pattern given the available features. We hence apply bias-correction (MBE +) to adjust the SVF^F at national scale based on data from the high-resolution DSM_{50cm}, similar to S_{sh} in Section 3.5.



Fig. 5. (a) Binary shadow maps used to derive S_{sh} , for 5 example time steps (0: cast shadow, 1: sun exposure). (b) Mean illumination map used to compute C_{sh} (red areas).

3.7. Flat rooftops

For a more realistic representation of PV panels on flat roofs, we assume PV panels to be installed in individual south-facing rows instead of placing them in consecutive rows along the roof aspect. Some spacing is left between the rows to reduce mutual shading effects between them (see Fig. 3). The tilt angle and row spacing is chosen as a trade-off between the total PV output and the capacity factor, which is a proxy of the economic feasibility. We have simulated different configurations and found a technically optimal trade-off for a tilt angle of 30° and a spacing of one panel height at the Swiss latitude.

The installation strategy on flat roofs impacts the PV potential in multiple ways: (i) tilted radiation increases due to the panel tilt of 30° , (ii) C_{pv} decreases due to the spacing of the rows, (iii) mutual shading between adjacent rows of panels increases the hourly shaded fraction and (iv) the sky view factor is reduced. The S_{sh} and the *SVF* of flat roofs are hence multiplied with an additional hourly shading factor and sky view factor, respectively, which we have simulated using a geometric model.

3.8. Uncertainty

The technical potential depends on multiple variables, as shown in Fig. 1. The best estimate for each variable is given by their first and second order moments, which may be used to combine various uncertainties using statistical methods. Uncertainties arise from different sources and are unknown in some cases. In this work, we consider only those uncertainties for which information can be extracted from the statistical analysis of our data. Further potential sources of uncertainty will be discussed in Section 5.4.

3.8.1. Uncertainty for ML methods

To estimate the uncertainties for the output variables of the ML models, namely G_h , G_B , ρ , C_{sh} and C_{pv} , we follow a two-stage approach. It distinguishes between the uncertainty arising from the modelling process (model uncertainty, σ_M) and the uncertainty related to the data noise (data uncertainty, σ_D) [60]. The data uncertainty may also represent errors introduced by previous processing steps, for example using GIS.

The model uncertainty is estimated as the standard deviation of the predictions from each ensemble member of the RF or ELM-E [55,60]. To quantify the data uncertainty, a second ML model is trained on the remaining residuals of the out-of-bag training data [54], which are derived from the squared difference between the targets and predictions [55,60]. This second ML model is used to predict σ_D for each predicted output. Further information regarding our implementation of this method is provided in [61]. The total uncertainty of a variable estimated using ML is then the squared sum of its model uncertainty and its data uncertainty.

3.8.2. Uncertainty for GIS methods

The uncertainty for the GIS-derived and bias corrected quantities $(S_{sh} \text{ and } SVF)$ is estimated from the residuals between the values computed using the national DSM_{2m} (S_{sh}^F, SVF^F) and the "true" values extracted from the DSM_{50cm} (S_{sh}^T, SVF^T) in Geneva. The error is treated as a random variable, whose first and second order moments are computed as the mean and the variance of these residuals. The mean is used for the bias correction, as explained in Sections 3.5 and 3.6, while the variance represents the uncertainty.

3.8.3. Uncertainty propagation

The propagation of uncertainty is performed by treating Eqs. (1)-(4) as functions of randomly distributed variables with statistical errors. To compute the variances of their output variables from a combination of the means, variances and covariances of their inputs (summarized in Table 4), we make the following assumptions: First, statistical

Table 9

Linear correlation coefficients for pairs of potentially correlated random variables. ¹denotes a spatio-temporal mean, ²indicates a mean across all time steps

	$G_h, G_B{}^1$	S_{sh} , SVF^2	G_t, A_{PV}^2	C_{pv}, C_{sh}
Correlation	0.87	0.36	0.04	0.02

independence is assumed between the solar radiation ($G_{h,B,D}$) and the DSM-derived variables (S_{sh} , SVF). This is valid as the uncertainty of $G_{h,B,D}$ is dominated by the meteorological variability, while that of $S_{sh}(t)$ and SVF is related to errors in the GIS methods. These are independent and uncorrelated processes. By contrast, G_h and G_B , as well as S_{sh} and SVF, exhibit a mutual correlation, as Table 9 shows. Therefore, their covariances must be considered in the uncertainty propagation. Second, we assume that G_t and A_{PV} , and C_{pv} and C_{sh} , are independent as their correlation coefficients are negligible (see Table 9). Third, we neglect the uncertainties of ρ and of the temperature data, as these have a low impact on the final results. Fourth, we do not account for the uncertainty related to the physical models of G_{Dt} , η_{PV} and PF (Section 3.1), due to a lack of data on their performance and the expected errors. This limitation will be discussed in Section 5.4.

Table 10 summarizes all variables for which uncertainties are considered, as well as the dimensions along which these are derived. Based on the above assumptions and given the uncertainties in Table 10, the variances of $G_D(\sigma_{GD}^2)$ and $G_t(\sigma_{Gl}^2)$ are derived from Eqs. (1) and (2) using error propagation theory [62,63], such that:

$$\sigma_{GD}^2 = \sigma_{Gh}^2 + \sigma_{GB}^2 - 2 \operatorname{Cov}(G_h, G_B)$$
(5)

$$\sigma_{Gt}^2 = \sigma_B^2 + \sigma_D^2 + \sigma_R^2 + 2 (\text{Cov}(B, D) + \text{Cov}(B, R) + \text{Cov}(D, R))$$
(6)

where the direct, diffuse and reflected components of G_t are denoted as $B = (1 - S_{sh}) * G_{Bt}$, $D = SVF * G_{Dt}$ and $R = G_{Rt}$. The expressions for their variances (σ_B^2 , σ_D^2 , σ_R^2) and covariances are provided in Appendix D.

The variances of A_{PV} (σ_A^2) and E_{PV} (σ_{PV}^2) are derived from Eqs. (3) and (4), respectively:

$$\sigma_A^2 = A_t^2 \left(\sigma_{Csh}^2 \ C_{pv}^2 + \sigma_{Cpv}^2 \ (1 - C_{sh})^2 + \sigma_{Csh}^2 \ \sigma_{Cpv}^2 \right)$$
(7)

$$\sigma_{PV}^2 = \eta_{PV}^2 \ PF^2 \ (\sigma_A^2 \ G_t^2 + \sigma_{Gt}^2 \ A_{PV}^2 + \sigma_{Gt}^2 \ \sigma_A^2) \tag{8}$$

4. Results

4.1. Physical potential estimation

The models of Section 3.3 are used to predict the MMH solar horizontal radiation and the monthly albedo for each pixel of the (200×200) m² output grid in Switzerland. The results, as well as the estimated model and data uncertainties (Section 3.8), are reported in Table 11 as monthly sums. The uncertainty of ρ is not shown as it is neglected in the uncertainty propagation. For G_h and G_B , the model uncertainty σ_M is negligible compared to the data uncertainty σ_D due to the large amount of training data. The model performs better for the prediction of G_h than G_B . This may be explained by the large relative data uncertainty of G_B (up to 32%), indicating that its hourly variability is higher than that of G_h . The results for ρ are close to the standard

Table 10

Summary of the uncertainties for each variable in the RPV potential estimation. The dimension (dim) of the uncertainties refers to R as roof surfaces and t as time steps.

	$G_{h,B,D}$	Ssh	SVF	Csh	Cpv	G_t	A_{PV}	E_{PV}
Uncertainty Dim. of σ	$\sigma_{Gh,B,D}$ $R \times t$	σ _{Ssh} t	σ_{SVF} 1	σ _{Csh} R	σ_{Cpv} R	$\frac{\sigma_{Gt}}{R \times t}$	σ_A R	σ_{PV} $R \times t$

Results for the estimation of G_h , G_B and G_D , showing the monthly predicted values (in kWh/m²) and the model (σ_M) and data (σ_D) uncertainties (no distinction for G_D), as percentage of the monthly radiation.

	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
G_h	44.9	66.1	114	145	168	180	183	151	115	77.1	45.5	36.6
$\sigma_M(\%)$	1.32	1.02	1.01	1.27	1.09	1.16	1.32	1.12	1.36	1.15	0.97	1.03
$\sigma_D(\%)$	19.1	15.8	15.7	13.8	13.6	11.8	13.7	12.9	19.9	18.4	19.0	14.0
G_B	22.1	35.2	62.3	79.6	87.6	101	107	89.7	68.2	43.1	23.3	17.8
$\sigma_M(\%)$	2.28	1.60	1.55	1.91	1.91	2.10	2.23	1.93	2.04	1.83	1.82	2.18
$\sigma_D(\%)$	30.1	23.1	26.8	26.2	29.2	23.2	26.0	22.5	32.9	26.3	31.8	26.2
G_D	22.8	30.9	51.8	65.3	80.1	79.6	75.4	60.9	47.0	34.0	22.2	18.8
$\sigma_{GD}(\%)$	19.0	17.0	17.3	16.1	15.8	14.6	18.3	16.7	24.7	20.9	19.3	13.5
ρ	0.52	0.50	0.41	0.32	0.26	0.22	0.19	0.19	0.22	0.27	0.36	0.48



Fig. 7. Spatial distribution of annual predicted G_h (a) and G_B (b), as well as the respective total uncertainties σ_{Gh} (c) and σ_{GB} (d).

value of 0.2 in the summer and reach up to 0.5 on average in the winter due to snow coverage at high altitudes in the Swiss alps.

Fig. 7a and b show the spatial distribution of G_h and G_B for Switzerland, while Fig. 7c and d show σ_{Gh} and σ_{GB} , which are dominated by the data uncertainty. The highest solar radiation and the largest uncertainty is found at high altitudes in the south of the country where weather extremes are frequent. The majority of buildings are however located at lower altitudes in the Swiss plateau, which spans from the north-west to the north-east of the country. The uncertainty tends to be low here, so the weather patterns are more predictable. In the plateau, also the values for ρ (not shown in Fig. 7) lie below the averages quoted in Table 11 with a low uncertainty, which shows that ρ mostly impacts the potential at high altitudes.

4.2. Geographic potential estimation

The available area for PV panel installation (A_{PV}) is obtained from the *shaded area coefficient* (C_{pv}) and the *panelled area coefficient* (C_{sh}) using Eq. (3). Its uncertainty σ_A is derived from C_{pv} , C_{sh} and their total uncertainties $(\sigma_{Cpv}, \sigma_{Csh})$ using Eq. (7).

Fig. 8 shows the final values for C_{pv} (a) and σ_{Cpv} (b) as a function of roof tilt and roof area, which is shown on a logarithmic scale. Large areas with a low tilt have the highest C_{pv} (70–80%) and a small uncertainty. This is due to the high number of installed panels, which reduces the relative effects of the roof shape and the presence of

superstructures on C_{pv} . Large roofs with a steep tilt have a high uncertainty, as they are rare. Due to the panel placement strategy for flat roofs (see Section 3.7), these tend to have a lower C_{pv} . The highest uncertainty appears for medium-sized roofs (10–100 m²) with C_{pv} in the range of 0.3–0.6. In this range, exact roof shapes as well as the favourable or unfavourable location of superstructures may change the number of installed panels considerably. Very small areas (<10 m²) have nearly zero available area and a low uncertainty, as the roof shape is frequently unsuitable for the installation of the rectangular panels.

The predicted values for $(1 - C_{sh})$, grouped by roof aspect and tilt angles, are shown in Fig. 9a, with the related σ_{Csh} shown in Fig. 9b. As expected, steep north-facing surfaces have the highest proportion of strongly shaded roof surface, i.e. the lowest values for $(1 - C_{sh})$, while this value is highest for steep south-facing surfaces. Interestingly, flat surfaces (in the centre of Fig. 9a) are significantly more shaded than roofs with a shallow tilt. This may be due to obstructing objects on flat surfaces or a stronger shading from surrounding buildings. The uncertainty is proportional to the shaded area coefficient and is hence highest for steep north-facing roofs.

To compute G_t from Eq. (2) and its uncertainty σ_{Gt} from Eq. (6), we combine the tilted radiation $G_{Bt,Dt,Rt}$ (Section 3.1) with the S_{sh} (Section 3.5) and the *SVF* (Section 3.6). Fig. 10a shows the bias-corrected unshaded fraction $(1 - S_{sh})$ for roofs of different tilts and aspects for four example hours in June. The patterns of S_{sh} follow the trajectory of the sun, with shading on west-facing surfaces in the morning hours and



Fig. 8. (a) Panelled area coefficient C_{pv} and (b) associated uncertainty σ_{Cpv} , for roofs with different tilt angle and roof area.

shaded east-facing roofs in the evening. Around solar noon (12 h–14 h), S_{sh} is nearly zero for all roofs. This is expected as strongly shaded areas are already excluded (as part of C_{sh}). Fig. 10b shows the related uncertainty σ_{Ssh} (blue line), which is strongly correlated with the zenith angle of the sun (orange line). As the uncertainty primarily arises from the discrepancy between the DSM_{2m} and the DSM_{50cm}, these results indicate that the discretization error of the DSM_{2m} has a higher impact on S_{sh} at low sun altitudes. Consequently, the uncertainty is high in the morning and evening and low during midday in the summer months, when the solar radiation is highest. The average *SVF* on Swiss roofs is 0.69, with a σ_{SVF} of 0.12 derived from the bias correction.

Fig. 11 shows the annual tilted irradiation on the building roofs (G_t) and its uncertainty (σ_{Gt}), again grouped by their tilt and aspect angles. The annual G_t (a) is highest for south-facing roofs with a tilt angle of around 40°. This matches the latitude of Switzerland. As panels on flat roofs are oriented south and tilted at 30°, they receive near-maximum solar irradiation. Fig. 11b shows σ_{Gt} as percentage of G_t . It is lowest for steep north-facing roofs, which receive low direct radiation and are hence less impacted by σ_{GB} . The highest relative uncertainty is found on steep roofs in the east and the west. These roofs receive their highest proportion of direct radiation at low sun positions, for which σ_{Syh} is high. Fig. 11c shows the hourly profiles for surfaces oriented north, east, west and south, averaged across all roofs of similar aspect. In summer, the peak of east-facing roofs in the morning is higher than that of the west-facing roofs in the afternoon. In winter, the opposite is the case. These results suggest that during summer the sky is clearer in the morning, while the weather conditions are generally better in the afternoon in winter.

4.3. Technical potential estimation

To obtain the technical PV potential (E_{PV}) and its uncertainty (σ_{PV}), Eqs. (4) and (8) are applied to G_t , A_{PV} , the module efficiency (η_{PV}) and the performance factor (*PF*), which are obtained as described in Section 3.1. Fig. 12 shows the annual E_{PV} for Switzerland, grouped by its three main characteristics: tilt, aspect and roof area. For comparability, the values are normalized by the total roof area (A_t). Fig. 12a shows the potential with respect to roof tilt (x-axis) and roof area (y-axis), indicating the highest potential for large roofs (>500 m²) with shallow tilt. The similarity to the pattern of C_{pv} in Fig. 8a demonstrates the strong dependency of E_{PV} on C_{pv} . As a consequence, the peak potential in Fig. 12b is no longer at 40° south, as it is the case for G_t (see Fig. 11), but it appears instead at 10°–20°, which is the tilt angle of many large roofs. Flat roofs, on the other hand, show a relatively smaller potential, due the selected strategy for installing PV panels on flat roofs (see Section 3.7).

To obtain a realistic large-scale potential estimate, we exclude roofs with a small available area of $A_{PV} < 8 \text{ m}^2$ [27], which represents a minimal economic feasibility [14,26,27,64]. We further exclude all north-facing roofs with an aspect angle $|\gamma| > 90^\circ$ from south [26,27], which aims at removing those roofs with a low PV potential. Other studies use a minimum annual G_t of 1000 kWh/m² [14,20,64]. This threshold however is found to be very sensitive to small changes in the estimated potential. The aspect angle is hence preferred as selection criterion.

Applying these criteria to all 9.6 million roof surfaces in Switzerland, 2.7 million roofs on 2.3 million buildings remain suitable to install PV panels. The suitable roofs represent 56.4% of the total A_{PV} of 267 ± 71 km², yielding a maximum technical PV potential of 24 ± 9 TWh. Fig. 13a shows the spatial distribution of the annual E_{PV} ,



Fig. 9. (a) Proportion of roof area suitable for PV installation based on shading effects $(1 - C_{sh})$, (b) its uncertainty σ_{Csh} per roof aspect and tilt.



Fig. 10. (a) Hourly unshaded fraction of suitable roof area $(1 - S_{sh})$ for four example hours in June, averaged across roofs with equal aspect and tilt angles and (b) its uncertainty σ_{Ssh} as a function of time. The yellow line shows the bias between the two DSMs, for which we correct the values of $1 - S_{sh}$.

aggregated to pixels of (500×500) m² for visualisation purposes. The potential is centered in the Swiss plateau, where most densely populated cities are located. The annual E_{PV} per rooftop of one such pixel is shown in Fig. 13b, which demonstrates that large flat roofs have by far the highest potential to install PV panels. Fig. 13c shows the temporal variation of the PV potential and its total uncertainty, in comparison with the Swiss electricity demand of 2018 [65]. In absolute terms, the uncertainty is highest during the summer months, due to the high variability of the horizontal radiation in these months. The RPV potential exceeds the electricity demand even for the lower boundary of E_{PV} ($E_{PV} - \sigma_{PV}$) from March until September, while a deficit is expected from November to January and during night hours.

4.4. Sensitivity analysis

We conduct a sensitivity analysis to quantify the impact of the individual parameters computed in this study on the technical PV potential. The analysis is performed in two stages: firstly, each parameter is independently varied in a fixed range of \pm 50%. Secondly, all parameters with a quantifiable uncertainty are varied by $\pm \sigma$. Due to the high correlation of the horizontal radiation components, G_h and G_B are varied simultaneously. A representative sample of 10,000 rooftops has been used for this analysis.

Fig. 14a shows the change in technical PV potential for varying each parameter by \pm 50%. The horizontal radiation components (G_h), and the fractions of available roof area (C_{sh} , C_{pv}) are the most sensitive parameters and thus exhibit the steepest slope. This is expected, as these



Fig. 11. (a) Annual tilted irradiation (G_t , in kWh/m²) and (b) relative annual uncertainty (σ_{Gt}/G_t) per roof aspect and tilt angle, (c) MMH profiles of G_t for north, east, west and south-facing roofs.



Fig. 12. Annual technical RPV potential for all roof surfaces (E_{PV} , in kWh/m²), (a) grouped by roof tilt and area, (b) grouped by roof tilt and aspect.

variables are directly correlated with the factors in the computation of E_{PV} (see Eq. (4)). As the panel efficiency reduces slightly for high G_h , its sensitivity curve lies just below C_{sh} , C_{pv} . Changes in S_{sh} and SVF have a smaller impact on the RPV potential, as they are multiplied with G_{Bt} and G_{Dt} , respectively. The curves flatten out for large positive changes, as both factors are saturating at a value of 1. The albedo (ρ) has a very low sensitivity, as there are few steep roofs that are impacted by a change in ρ . The ambient temperature (T_{amb}) is the only curve with a negative slope, as high temperatures decrease η_{PV} . Its sensitivity is rather low, as T_{amb} only indirectly impacts the PV potential as part of the physical model of η_{PV} .

Fig. 14b shows the impact of varying each variable within their

uncertainty ($\pm \sigma$). The upper and lower dashed lines represent the propagated σ_{PV} . The results suggests that in a year with low G_h , the electricity generation may be up to 18% lower than estimated, and similarly for the higher case. The S_{sh} and SVF show the lowest sensitivity, which agrees with Fig. 14a. Comparing C_{pv} and C_{sh} shows that σ_{Cpv} , driven by the high uncertainty of medium-sized roofs, is overall larger than σ_{Csh} , with a potential impact of \pm 11%. The bars in Fig. 14b indicate the potential change when all roofs are considered, while the lines show the potential change for the suitable roofs only (see Section 4.3). For G_t , the suitable south-facing roofs have a higher uncertainty than all roofs, as expected from Fig. 11b. The opposite is the case for A_{PV} , as the suitable roofs have a higher proportion of flat roofs with a low σ_{Cpv} .



Fig. 13. (a) Spatial distribution of annual E_{PV} , aggregated to pixels of (500 × 500) m² for visualization purposes, (b) annual E_{PV} for the suitable roofs of a randomly selected (500 × 500) m² pixel in the city of Geneva, (c) monthly-mean-hourly profiles of E_{PV} , summed for all suitable roofs, the σ_{PV} and the Swiss electricity demand of 2018.



Fig. 14. Change in RPV potential for the variation of (a) each variable by \pm 50% and (b) each uncertain variable by $\pm \sigma$. Considered variables are the horizontal radiation (G_h), partly shading factor (S_{sh}), sky view factor (*SVF*), surface albedo (ρ), shaded and panelled area ratios (C_{sh} , C_{pv}) and ambient temperature (T_{amb}).

5. Discussion

5.1. Methodological contribution

Our methodology presents an end-to-end approach to estimate the RPV potential at monthly-mean-hourly temporal resolution for individual roof surfaces. It uses large building and environmental datasets and may be transferred to any region or country where such data is or will become available. The proposed method adapts best existing practices for a data-driven estimation of each parameter that impacts RPV potential. This includes (i) the spatio-temporal variation of the horizontal solar radiation, (ii) the effects of surrounding trees and buildings on roof shading and the sky view factor, (iii) the impact of roof geometry and superstructures on the available area for installing PV panels, and (iv) the temperature-dependence of the PV module efficiency. As a result, the potential computed here is more accurate and provides a higher spatio-temporal resolution than previous works.

The strength of our method lies in the combination of physical models with GIS and ML. The former two provide a detailed representation of the physical processes underlying the RPV assessment. The use of ML, applied previously only in [26,27], allows to include additional knowledge extracted from data with only partial spatial or temporal coverage, and hence improves the accuracy of the results. The ML approach also outlines a method to use sparsely available datasets for a PV assessment in other regions.

Furthermore, we propose a structured method to estimate and propagate uncertainty in large-scale PV potential studies. Using the ML and GIS methods, we are able to quantify uncertainties related to the data sources and individual processing steps. These values are then combined using statistical tools to estimate the total uncertainty. This allows decision-makers to better understand the potential contribution of RPV to future energy systems.

5.2. Practical contribution

The application of the methodology to Switzerland provides several

practical contributions for an effective integration of RPV in the built environment. Firstly, our results form a dataset of hourly profiles of RPV potential and its uncertainty for 9.6 M rooftops in Switzerland, which will be publicly accessible. To the best of our knowledge, it is the first dataset of its type at this spatio-temporal resolution and scale. Our data can be used by the research community to study future energy systems with large shares of RPV, whereby the uncertainty permits the modelling of different scenarios.

Secondly, the large-scale estimate of 24 ± 9 TWh provides a realistic estimate of the maximum technical RPV potential for Switzerland. According to our data, electricity generation from PV may cover over 40% of Switzerland's annual demand of 57.6 TWh in 2018 [66]. However, our results also show that the potential contribution from RPV is insufficient during winter and night hours, while there may be a large surplus of RPV generation during peak hours in summer (see Fig. 13b).

Thirdly, the sensitivity analysis performed here identifies the parameters with the highest impact on the estimated RPV potential, as well as the main sources of uncertainty (see Fig. 14). The parameter with the highest impact is the horizontal radiation. Its uncertainty is mainly caused by the high intermittency of the solar resource and can not be significantly reduced, even if higher-quality data was available. The panelled area ratio represents the second largest source of uncertainty. It is due to inaccuracies in the input data and uncertainty in the modelling approach. These may be reduced by a more precise model, for example using image processing techniques [24], or a more detailed building model (i.e. LOD 4) that is available in the entire study region.

5.3. Comparison with existing studies

To set our large-scale estimate into context, we compare it to five national studies on RPV potential in Switzerland [20,22,26–28], including a national project known as *Sonnendach/toitsolaire* [22,64]. This allows to validate the magnitude of our results against existing work and to point out the improvements achieved through our methodology.

Table 12

Comparison of the results presented in 6 studies of technical RPV potential in Switzerland. To obtain comparable results, the entries labelled with * are computed from values quoted in the respective publications as explained in [67].

Study	A_{PV} (km ²)	Suitable roofs (%)	G_t (kWh/m ²)	η _{sys} (%)	E_{PV} (TWh)
IEA [28]	251*	55	1,088*	10	15.04
Assouline et al. [26]	328	60.5*	662*	13.6	17.86
Assouline et al. [27]	252	60.5*	786*	13.6	16.29
Sonnendach [22,64]	439*	71.6*	1243*	13.6	53.09*
Buffat et al. [20]	485	70.1*	1176*	10.3	41.20*
Present study	267	56.4	1186	13.8	24.58

Table 12 shows a quantitative comparison of all studies, which is further detailed in [67]. The metrics used for comparison are the total available area (A_{PV}), the percentage of total roof surface that is suitable for installing PV, the annual tilted irradiation (G_t), the system efficiency (η_{sys}), which combines module efficiency and performance factor, and the annual PV potential (E_{PV}).

Our results lie in the mid-range of the existing work. The estimate for A_{PV} is relatively low in comparison with other studies. It is in the range of the estimates by Assouline et al. [26,27], which use similar data-driven methods for estimating A_{PV} as applied here. This suggests that the ratio of available roof area for PV installation lies much below current expert recommendations, which are used in [64]. This may be due to the fact that current recommendations are focused on roofs with a high potential, which may not be representative for all roofs in the Swiss building stock. In [20], the available area for PV is not specifically addressed.

The annual tilted irradiation is relatively high and comparable to that estimated in [20,22]. Both studies apply best practices for the estimation of shading effects and use high-resolution satellite data. A validation of G_t against measurement data is provided in [20], who find a negligible mean error in summer when production is highest, and a small overestimation in winter. This suggests that the annual irradiation is close to its real value, while it is likely underestimated in [26–28], possibly due to a different computation of the shading effects.

Several complementarities exist between this study and previous work, given by the use of similar datasets and methods. This leads to comparable aggregated values and allows for the validation of our results against existing approaches. The added value of our work is given firstly by the computation of the results in monthly-mean-hourly resolution, instead of yielding monthly or yearly values as used in [22,26,27]. Secondly, we contribute to the existing work by assessing the available area for PV installation for each roof surface, rather than using communes [26] or pixel sizes of (200×200) m² [27]. Thirdly, our study is the only one that quantifies the uncertainty on the final potential estimate. Uncertainties are not specifically addressed in [26–28] and qualitatively assessed in [20,22].

5.4. Limitations

As all large-scale potential studies, our study relies on various data sources and a combination of statistical and empirical modelling steps. In this work, emphasis has been placed on systematically identifying and combining uncertainties in order to obtain a global uncertainty estimate for the technical RPV potential. Furthermore, a sensitivity analysis has been conducted to assess the impact of the uncertainty of individual steps on the final potential.

Some assumptions were made that impact these results: (i) The input data is assumed to be coherent and without error. In reality, some features may be missing or incorrectly represented, and discrepancies between the datasets may arise from different data collection dates and methods of creation. (ii) The systematic error introduced by using a fixed number of azimuth angles for computing the shading effects (38 bins) and the SVF (32 bins) is neglected. This error is expected to be negligible as the number of bins used here is relatively high. (iii) The potential uncertainties arising from the physical models of tilted radiation and module efficiency are neglected, as no comprehensive quantification of these has been found in the literature. (iv) The PV panel performance parameters as well as the system losses are treated as assumptions rather than random variables, due to the large variety of available technologies and their fast development. (v) PV panels on flat roofs are all placed in a technically optimal fashion (see Section 3.7). A comparison with other installation practices, such as lower tilt angles or the alternation of east and west-facing rows, is beyond the scope of this work. To consider different configurations of PV on flat roofs and the uncertainty in technological developments, a scenario-based approach may be followed.

5.5. Application and future work

A monthly-mean-hourly technical RPV potential for Switzerland is useful for applications including policy making, urban planning and the design of future energy systems. Policy makers may use our results, aggregated at regional or national scale, in order to formulate effective policies to integrate RPV into the built environment. Urban planners may assess the potential self-consumption and the electricity demand which could be covered by installing PV on existing roofs. They may further estimate the expected PV yield for new roofs taking into account the roof size, tilt and orientation from our results. Energy system designers can use this work to simulate future electricity networks at local scale, which allows to assess the potential mismatch between supply and demand and the resulting storage needs.

The generated dataset may be further used to estimate the PV potential for future scenarios that account for urbanization, climate change and technological advancement of PV. This is possible as the existing building stock covers different types of roofs in various urban contexts and climatic conditions, which exist in the Swiss plateau, the Jura mountains and the Alps. A techno-economic potential may be formulated for these scenarios by combining the maximum technical potential with economic factors such as installation and operational cost.

Finally, a further validation of the estimated RPV potential provides a challenging subject of future work. We validate our results against those of previous work (Section 5.3), some of which have been compared to measurement data [20]. While this approach is feasible for validating the tilted radiation, a direct validation of the available area is difficult due to the lack of a "ground truth". A possible method to establish such a ground truth may be the application of image segmentation techniques, for example using Convolutional Neural Networks [68], to high-resolution aerial imagery. This would allow for a more accurate detection of obstructing objects on rooftops as well as the recognition of already installed PV panels.

6. Conclusion

In this work, we present a big data mining approach to estimate the PV potential on 9.6 million rooftops at monthly-mean-hourly temporal resolution and propose a quantification the uncertainty on the estimated potential. The developed Machine Learning methodology uses high-resolution building and environmental data to extract information from data which is only available in parts of the study area. The Machine Learning algorithms are further used to quantify the uncertainties related to the estimated parameters. These are combined with uncertainties arising from other modelling steps and propagated to obtain an uncertainty on the technical PV potential estimate.

The national-scale application of our method results in a total rooftop PV potential for Switzerland of 24 ± 9 TWh, which could meet more than 40% of the country's electricity demand in 2018. This potential is in the mid-range of other large-scale estimates for Switzerland. The added value of our approach lies in the higher spatio-temporal resolution of the resulting datasets compared to existing studies, as well as in the uncertainty estimate for each spatio-temporal instance. This highlights areas and time spans with potentially large inaccuracies. Our results may be used to quantify the potential mismatch between PV production and supply, which is relevant for the design of future energy efficient districts and for the formulation of regional and national incentive policies for the diffusion of rooftop PV.

The work presented in this study provides an important contribution for the decarbonisation policies in Switzerland, as it enables the largescale modelling of future electricity grids with high shares of rooftop PV using hourly data for individual buildings. The proposed PV assessment method using data mining and uncertainty propagation is transferable to any region or country with sufficient high-quality data, where it can contribute to the transition towards low-carbon energy systems.

(A.2)

reviews and helpful comments on an earlier version of this article that significantly improved the final version of the paper. This research has

been financed partly by the Swiss National Science Foundation (SNSF)

under the National Research Program 75 (Big Data) for the HyEnergy

project, No. 407540_1167285, and partly by the Swiss Innovation Agency Innosuisse under the Swiss Competence Center for Energy

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

We thank the reviewers, as well as the editor for very thorough

Appendix A. Physical models

A.1. Tilted radiation

Widely accepted physical and empirical models are used to compute the plane-of-array (POA) radiation. The G_{Bt} is computed from G_B and the angle of incidence of the sun rays on the tilted panel (θ) [69]:

Research SCCER FEEB&D.

$$G_{Bt} = G_B * max \left(0, \frac{\cos(\theta)}{\cos(\theta_Z)}\right)$$
(A.1)

 $\cos(\theta) = \sin(\beta)\sin(\theta_Z)\cos(\gamma_S - \gamma) + \cos(\beta)\cos(\theta_Z)$

The angles θ_Z and γ_S describe the sun zenith and azimuth angles, respectively, while roof tilt and aspect are given by β and γ . The empirical formula for G_{Dt} using the Perez model is given by [47]:

$$G_{Dt} = G_D * \left[(1 - F_1) \left(\frac{1 + \cos\beta}{2} \right) + F_1 \frac{a}{b} + F_2 \sin\beta \right]$$
(A.3)

where F_1 and F_2 are empirically fitted functions for the circumsolar and horizon brightness, and *a*, *b* are geometric angles. The derivation of these factors is described in [46]. The calculation of G_{Rt} from the surface albedo ρ is given by [70]:

$$G_{Rt} = G_h * \rho \left(\frac{1 - \cos\beta}{2}\right) \tag{A.4}$$

To ease the notation, Eqs. (A.1)-(A.4) can be referred to as:

$$G_{Bt} = F_B G_B, \quad G_{Dt} = F_D G_D, \quad G_{Rt} = F_R G_h \tag{A.5}$$

A.2. Module and inverter efficiency

The module and inverter efficiencies are computed using the *PVWatts* model developed by the National Renewable Energy Laboratory (NREL) [50]. The DC power output of a PV panel is modelled as a function of the tilted radiation G_t and the cell temperature T_{cell} :

$$P_{dc} = \frac{G_t}{1000} P_{dc0} \left(1 + \gamma_{pdc} (T_{cell} - T_{ref})\right)$$
(A.6)

where P_{dc0} is the DC rating of the panel, γ_{pdc} is its temperature coefficient and T_{ref} is the reference temperature. We use average panel specifications of mid-range 60-cell mono-crystalline PV modules, the most frequently used technology in Switzerland [20], from three market-leading manufacturers (JA Solar, Jinko Solar, Trinasolar). The *PVSyst* model [71] is used to derive T_{cell} from G_t and the ambient temperature T_{amb} :

$$T_{cell} = T_{amb} + G_t \frac{\alpha (1 - \eta_m)}{U}$$
(A.7)

where α denoted the absorption coefficient, η_m denotes the module efficiency and *U* is the heat transfer component. From the DC power output and the area of a PV panel (A_{panel}), we compute the module efficiency η_{PV} :

$$\eta_{PV} = \frac{P_{dc}}{G_t * A_{panel}} \tag{A.8}$$

The empirical loss formula for the inverter efficiency (η_{inv}) in *PVWatts* is given by:

$$\eta_{inv} = \frac{\eta_{nom}}{\eta_{ref}} \left(-0.0162 * \zeta - \frac{0.0059}{\zeta} + 0.9858 \right)$$
(A.9)

where $\zeta = P_{dc}/P_{dc0}$, η_{nom} is the nominal inverter efficiency and η_{ref} is the reference efficiency. The performance factor (*PF*) is obtained by multiplying η_{inv} with other system losses. These include soiling, degradation, mismatch, wiring and connection losses and are estimated as 14% [50].

We use the pulib python package developed by Sandia National Laboratories [72] for the computing the POA irradiance and the module and inverter efficiency. All parameter values are shown in Table A.13.

Table A.13	
Parameters used in the PV module and inverter efficiency models [50,	71].

Parameter	Value	Description
P_{dc0}	285 Wp	Nameplate DC rating
γ_{pdc}	− 0.39 %/°C	Temperature coefficient
Tref	25 °C	Cell reference temperature
α	0.9	Absorption coefficient
η_m	0.17	Nominal module efficiency
U	15 W/m ² K	Heat transfer component
Apanel	1.6 m ²	PV panel area
η_{nom}	0.96	Nominal inverter efficiency
η_{ref}	0.9637	Reference inverter efficiency

Appendix B. Tuning of Machine Learning models

The tuning of the ML models aims at selecting the architectural parameters (hyper-parameters) which optimize the model performance. The mean-squared error (MSE) is used as means to assess performance and is evaluated on a validation set. This is a random subset of the training data (here 20%), which is excluded from the training procedure. Both Machine Learning algorithms (ELM-E and RF) are implemented in python. The RF uses the scikit-learn [58] library and the ELM-E is developed based on the HPELM package [73] with GPU acceleration.

B.1. Extreme Learning Machine Ensemble

To estimate G_h and G_B (Section 3.3), we use training data from 11,243 pixel coordinates in Switzerland and for 4,663 h with non-zero solar radiation, which are split into 12 monthly subsets. The training data for ρ are 365 daily values for each coordinate. The output grid of (200×200) m² pixels results in a prediction set of 1.04 M locations, with 146 non-zero MMH time steps (for G_h , G_B) or 12 monthly values (for ρ). As Extreme Learning Machines (ELM) are single-layer neural networks, the ELM-E has two hyper-parameters that are tuned, (i) the number of nodes of the hidden layer of each neural network m, and (ii) the number of ELM in the ensemble n. A large m may lead to over-fitting, while a small m may result in a poor performance. Table B.14 shows the trade-off between the model performance (validation MSE) and the computational time for training and prediction (on the validation set), which exists in the choice of m. The given data is for G_h in June, but similar trends exist for the other months, for G_B and ρ . The tuning for n is not shown, as it is inhibited only by computational time. The selected hyper-parameters are m = 1000, n = 50 for G_h and G_B and m = 1000, n = 30 for ρ .

Table B.14

Results of the tuning of m, showing the trade-off between the validation MSE and training and prediction times (per ELM), for G_h in June. The selected value for m is shown in bold.

m	50	100	500	1000	2000	5000	10000
Validation MSE	0.084	0.052	0.044	0.043	0.043	0.042	0.042
Training time (s)	16	18	18	23	43	230	1040
Prediction time (s)	2.5	3.0	2.7	3.0	3.4	5.4	7.8

B.2. Random Forest

Random Forests are used for two separate models, to estimate C_{pv} (Section 3.4) and C_{sh} (Section 3.5). Due to the smaller size of the data as compared to the ELM-E, a 5-fold cross-validation is applied. It splits the data into 5 subsets, from which 5 separate models are trained and validated each on a different subset. The RF has three main hyper-parameters which can be tuned: the number of trees in the ensemble (*n*), the minimum number of samples in each leaf node (*l*), and the number of features (*m*) which are randomly considered for splitting at each node. *m* determines the level of randomization within each decision tree, *l* impacts the level of smoothing of the predictions, while *n* improves the robustness of the model, and is primarily limited by computational time [51]. The selected hyper-parameters for both models are *l* = 3, *m* = 5 and *n* = 1000. It should be noted that the RF is much less prone to small deviations in the model hyper-parameters than the ELM-E.

Appendix C. Geospatial algorithms

C.1. Virtual installation of PV panels

Algorithm 1 details the geospatial algorithm that is implemented to virtually install PV panels on roof surfaces. It uses the python geopandas library [74]. The output of the geospatial algorithm, C_{pv} , is calculated for each roof surface as the ratio between the roof area covered by N_{panels} virtual panels, each of area A_{panel} , and the tilted area A_t of the roof surfaces:

$$C_{pv} = \frac{N_{panels} \times A_{panel}}{A_t} \tag{C.1}$$

Algorithm 1. PV panel placement on rooftop polygons

- 1: Rotate roof surfaces to face south
- 2: Create inside buffers (0.4 m) along the roof edges
- 3: Project PV panel dimensions for each tilted roof
- for all roof surfaces:
- 4: Place rectangular grid on south-facing roof
 5: Remove grid cells (panels) outside roof surface
- Remove grid cells (panels) outside roof surf
 Count panels and rotate to roof aspect
- 7: Compute panelled area and panelled area ratio

C.2. Shading effects

The shading effects, namely S_{sh} and C_{sh} , are computed using horizon maps for each azimuth angle of the sun. Algorithm 2 describes the steps to extract S_{sh} and C_{sh} from the horizon maps. The computation of these maps is the most computationally expensive step in the estimation of hourly RPV potential. Hence, several measures are taken to assure the feasibility of the approach:

- 1. Horizon maps are computed in bins of 5° for south-east to south-west sun azimuths (110–250°), and in bins of 10° for east (60–110°) and west (250–300°) azimuths, where solar radiation is low. Thus, only 38 maps need to be computed.
- 2. The horizon distance is 100 m, i.e. a radius of 100 m is considered around each pixel to compute horizons. In Switzerland, where few high-rise buildings exist, this threshold is suitable to estimate the shading effects from surrounding trees and buildings.
- 3. The algorithm is implemented using the GRASS GIS engine [75]. The Swiss terrain is split into 16 sub-regions and processed in parallel to reduce the computational time from ~1000 to ~30 h.

Algorithm 2. Computation of shading effects on rooftops

- 1: Compute horizon maps from DSM for 38 angle bins
- 2: Crop horizon maps over (rasterized) roofs
 - for all t:
- 3: Apply binary filter to horizon maps using the solar position
- 4: Average binary shading masks over t (sun exposure map)
- 5: Extract roof areas with sun exposure <40%
- 6: Compute shaded area coefficient C_{sh}
- for all t:
- 7: Extract roof areas with sun exposure >40%
- 8: Compute hourly shading fraction $S_{sh}(t)$

C.3. Sky view factor

The sky view factor is computed from similar horizon maps as described above for 32 equally spaced azimuth bins and a horizon distance of 100 m, using the *skyview* add-on for GRASS GIS [59]. The computation is performed in parallel over 64 sub-regions.

Appendix D. Uncertainty propagation for tilted radiation

The variance and covariance of the three addends in Eq. (2) (labelled *B*, *D*, *R* in Eq. (6)) are computed from the mean and variance of the horizontal radiation ($G_{B,D,h}$, $\sigma_{GB,GD,Gh}^2$), the partly shaded factor ($(1 - S_{sh})$, σ_{sh}^2) and the sky view factor (*SVF*, σ_{SVF}^2) such that:

$$\begin{split} \sigma_B^2 &= F_B^2(\sigma_{Ssh}^2 \sigma_B^2 + \sigma_{GB}^2(1 - S_{sh})^2 + \sigma_{Ssh}^2 \sigma_{GB}^2) \\ \sigma_D^2 &= F_D^2(\sigma_{SVF}^2 G_D^2 + \sigma_{GD}^2 SVF^2 + \sigma_{SVF}^2 \sigma_{GD}^2) \\ \sigma_R^2 &= F_R^2 \sigma_{Gh}^2 \end{split}$$

where F_B , F_D , F_R are the beam, diffusion and reflection factors introduced in Eq. A.5. Under the assumptions in Section 3.8.3, we derive the covariances between the beam, diffuse and reflected components of G_t as a function of the covariances between the horizontal radiation components, as well as the covariance between the shading and the SVF, such that:

$$Cov(B, D) = F_B F_D \times (Cov(G_B, G_D) \times Cov(S_{sh}, SVF) + G_B G_D \times Cov(S_{sh}, SVF) + (1 - S_{sh}) \times SVF \times Cov(G_B, G_D))$$

$$Cov(B, R) = F_B F_R \times (1 - S_{sh}) \times Cov(G_B, G_h)$$

$$Cov(D, R) = F_D F_R \times SVF \times Cov(G_D, G_h)$$
(D.2)

where the covariances $Cov(G_h, G_D)$ and $Cov(G_B, G_D)$ are computed from σ_{Gh} , σ_{GB} and σ_{GD} in analogy to Eq. (5).

Appendix E. Supplementary material

The data associated with this article will be published in Zenodo under the following DOI: 10.5281/zenodo.3609833.

(D.1)

References

- Peters GP, Andrew RM, Canadell JG, Fuss S, Jackson RB, Korsbakken JI, et al. Key indicators to track current progress and future ambition of the Paris Agreement. Nat Climate Change 2017;7:118–22. https://doi.org/10.1038/nclimate3202.
- [2] Jäger-Waldau A. European Commission. Joint Research Centre, PV status report 2018; 2018. OCLC: 1111219935.
- [3] Bódis K, Kougias I, Jäger-Waldau A, Taylor N, Szabó S. A high-resolution geospatial assessment of the rooftop solar photovoltaic potential in the European Union. Renew Sustain Energy Rev 2019;114:109309. https://doi.org/10.1016/j.rser.2019. 109309.
- [4] Levinson R, Akbari H, Pomerantz M, Gupta S. Solar access of residential rooftops in four California cities. Sol Energy 2009;83:2120–35. https://doi.org/10.1016/j. solener.2009.07.016.
- [5] Nguyen HT, Pearce JM. Incorporating shading losses in solar photovoltaic potential assessment at the municipal scale. Sol Energy 2012;86:1245–60. https://doi.org/ 10.1016/j.solener.2012.01.017.
- [6] Santos T, Gomes N, Freire S, Brito MC, Santos L, Tenedório JA. Applications of solar mapping in the urban environment. Appl Geogr 2014;51:48–57. https://doi.org/10. 1016/j.apgeog.2014.03.008.
- [7] Strzalka A, Alam N, Duminil E, Coors V, Eicker U. Large scale integration of photovoltaics in cities. Appl Energy 2012;93:413–21. https://doi.org/10.1016/j. apenergy.2011.12.033.
- [8] Phillips C, Elmore R, Melius J, Gagnon P, Margolis R. A data mining approach to estimating rooftop photovoltaic potential in the US. J Appl Stat 2019;46:385–94. https://doi.org/10.1080/02664763.2018.1492525.
- [9] Kodysh JB, Omitaomu OA, Bhaduri BL, Neish BS. Methodology for estimating solar potential on multiple building rooftops for photovoltaic systems. Sustain Cities Soc 2013;8:31–41. https://doi.org/10.1016/j.scs.2013.01.002.
- [10] Schallenberg-Rodríguez J. Photovoltaic techno-economical potential on roofs in regions and islands: the case of the Canary Islands. Methodological review and methodology proposal. Renew Sustain Energy Rev 2013;20:219–39. https://doi. org/10.1016/j.rser.2012.11.078.
- [11] Khan MMA, Asif M, Stach E. Rooftop PV potential in the residential sector of the Kingdom of Saudi Arabia. Buildings 2017;7:46. https://doi.org/10.3390/ buildings7020046.
- [12] Ordóñez J, Jadraque E, Alegre J, Martínez G. Analysis of the photovoltaic solar energy capacity of residential rooftops in Andalusia (Spain). Renew Sustain Energy Rev 2010;14:2122–30. https://doi.org/10.1016/j.rser.2010.01.001.
- [13] Mansouri Kouhestani F, Byrne J, Johnson D, Spencer L, Hazendonk P, Brown B. Evaluating solar energy technical and economic potential on rooftops in an urban setting: the city of Lethbridge, Canada. Int J Energy Environ Eng 2019;10:13–32. https://doi.org/10.1007/s40095-018-0289-1.
- [14] Romero Rodríguez L, Duminil E, Sánchez Ramos J, Eicker U. Assessment of the photovoltaic potential at urban level based on 3d city models: a case study and new methodological approach. Sol Energy 2017;146:264–75. https://doi.org/10.1016/j. solener.2017.02.043.
- [15] Hong T, Lee M, Koo C, Jeong K, Kim J. Development of a method for estimating the rooftop solar photovoltaic (PV) potential by analyzing the available rooftop area using Hillshade analysis. Appl Energy 2017;194:320–32. https://doi.org/10.1016/ j.apenergy.2016.07.001.
- [16] Singh R, Banerjee R. Estimation of rooftop solar photovoltaic potential of a city. Sol Energy 2015;115:589–602. https://doi.org/10.1016/j.solener.2015.03.016.
- [17] Jakubiec JA, Reinhart CF. A method for predicting city-wide electricity gains from photovoltaic panels based on LiDAR and GIS data combined with hourly Daysim simulations. Sol Energy 2013;93:127–43. https://doi.org/10.1016/j.solener.2013. 03.022.
- [18] Lukac N, Seme S, Žlaus D, Štumberger G, Žalik B. Buildings roofs photovoltaic potential assessment based on LiDAR (Light Detection And Ranging) data. Energy 2014;66:598–609. https://doi.org/10.1016/j.energy.2013.12.066.
- [19] Wegertseder P, Lund P, Mikkola J, García Alvarado R. Combining solar resource mapping and energy system integration methods for realistic valuation of urban solar energy potential. Sol Energy 2016;135:325–36. https://doi.org/10.1016/j. solener.2016.05.061.
- [20] Buffat R, Grassi S, Raubal M. A scalable method for estimating rooftop solar irradiation potential over large regions. Appl Energy 2018;216:389–401. https://doi. org/10.1016/j.apenergy.2018.02.008.
- [21] Ramirez Camargo L, Zink R, Dorner W, Stoeglehner G. Spatio-temporal modeling of roof-top photovoltaic panels for improved technical potential assessment and electricity peak load offsetting at the municipal scale. Comput Environ Urban Syst 2015;52:58–69. https://doi.org/10.1016/j.compenvurbsys.2015.03.002.
- [22] Klauser D. Solarpotentialanalyse f
 ür Sonnendach.ch, Schlussbericht, Bundesamt f
 ür. Energie BFE 2016.
- [23] Desthieux G, Carneiro C, Camponovo R, Ineichen P, Morello E, Boulmier A, et al. Solar energy potential assessment on rooftops and facades in large built environments based on LiDAR data, image processing, and cloud computing. Methodological background, application, and validation in Geneva (Solar Cadaster) Front Built Environ 2018*4. https://doi.org/10.3389.(buil 2018.00014)
- Cadaster). Front Built Environ 2018;4. https://doi.org/10.3389/fbuil.2018.00014.
 [24] Mainzer K, Killinger S, McKenna R, Fichtner W. Assessment of rooftop photovoltaic potentials at the urban level using publicly available geodata and image recognition techniques. Sol Energy 2017;155:561–73. https://doi.org/10.1016/j.solener.2017. 06.065.
- [25] Palmer D, Koumpli E, Cole I, Gottschalg R, Betts T. A GIS-based method for identification of wide area rooftop suitability for minimum size PV systems using LiDAR data and photogrammetry. Energies 2018;11:3506. https://doi.org/10.3390/

en11123506.

- [26] Assouline D, Mohajeri N, Scartezzini J-L. Quantifying rooftop photovoltaic solar energy potential: a machine learning approach. Sol Energy 2017;141:278–96. https://doi.org/10.1016/j.solener.2016.11.045.
- [27] Assouline D, Mohajeri N, Scartezzini J-L. Large-scale rooftop solar photovoltaic technical potential estimation using Random Forests. Appl Energy 2018;217:189–211. https://doi.org/10.1016/j.apenergy.2018.02.118.
- [28] IEA. Potential for Building Integrated Photovoltaics, Technical Report IEA PVPS T7-4. International Energy Agency (IEA); 2002.
- [29] Wiginton LK, Nguyen HT, Pearce JM. Quantifying rooftop solar photovoltaic potential for regional renewable energy policy. Comput Environ Urban Syst 2010;34:345–57. https://doi.org/10.1016/j.compenvurbsys.2010.01.001.
- [30] Thevenard D, Pelland S. Estimating the uncertainty in long-term photovoltaic yield predictions. Sol Energy 2013;91:432–45. https://doi.org/10.1016/j.solener.2011. 05.006.
- [31] Gueymard CA. Direct and indirect uncertainties in the prediction of tilted irradiance for solar engineering applications. Sol Energy 2009;83:432–44. https://doi.org/10. 1016/j.solener.2008.11.004.
- [32] Kreifels N, Killinger S, Fischer D, Wille-Haussmann B. Uncertainty and error analysis of calculation procedures for PV self-consumption and its significance to investment decisions. In: 2016 13th international conference on the European Energy Market (EEM); 2016. p. 1–5. https://doi.org/10.1109/EEM.2016.7521221. iSSN: 2165-4093.
- [33] Martins TAdL, Adolphe L, Bastos LEG, Martins MAdL. Sensitivity analysis of urban morphology factors regarding solar energy potential of buildings in a Brazilian tropical context. Sol Energy 2016;137:11–24. https://doi.org/10.1016/j.solener. 2016.07.053.
- [34] Izquierdo S, Rodrigues M, Fueyo N. A method for estimating the geographical distribution of the available roof surface area for large-scale photovoltaic energypotential evaluations. Sol Energy 2008;82:929–39. https://doi.org/10.1016/j. solener.2008.03.007.
- [35] Stöckli R. The HelioMont Surface Solar Radiation Processing (2017 Update). Technical Report 93. MeteoSwiss; 2017.
- [36] Ineichen P. Long term satellite global, beam and diffuse irradiance validation. Energy Procedia 2014;48:1586–96. https://doi.org/10.1016/j.egypro.2014.02. 179.
- [37] Frei C. Interpolation of temperature in a mountainous region using nonlinear profiles and non-Euclidean distances. Int J Climatol 2014;34:1585–605. https://doi. org/10.1002/joc.3786.
- [38] Stöckli R. Daily, monthly and yearly satellite-based global radiation. Documentation of MeteoSwiss Grid-Data Products, Federal Office of Meteorology and Climatology MeteoSwiss; 2013.
- [39] MeteoSwiss. Daily mean, minimum and maximum temperature. Documentation of MeteoSwiss Grid-Data Products, Federal Office of Meteorology and Climatology MeteoSwiss; 2017.
- [40] Federal Statistics Office, Federal Register of Buildings and Dwellings; 2015. URL https://www.bfs.admin.ch/bfs/en/home/register/gebaeude-wohnungsregister. html.
- [41] SITG. Superstructures des toits des bâtiments; 2019. https://ge.ch/sitg/fiche/1148.
- [42] Robert S, Foresti L, Kanevski M. Spatial prediction of monthly wind speeds in complex terrain with adaptive general regression neural networks. Int J Climatol 2012;33:1793–804. https://doi.org/10.1002/joc.3550.
- [43] Swisstopo. swissALTI3d The high precision digital elevation model of Switzerland; 2017. URL https://shop.swisstopo.admin.ch/en/products/height_models/alti3D.
- [44] Swisstopo. DSM The digital surface model of Switzerland, 2005. URL https://shop. swisstopo.admin.ch/en/products/height_models/DOM.
- [45] SITG. MNS modèle numérique de surface 2013 (pixel 0.5 m); 2018. https://ge.ch/ sitg/fiche/9330.
- [46] Loutzenhiser PG, Manz H, Felsmann C, Strachan PA, Frank T, Maxwell GM. Empirical validation of models to compute solar irradiance on inclined surfaces for building energy simulation. Sol Energy 2007;81:254–67. https://doi.org/10.1016/ j.solener.2006.03.009.
- [47] Perez R, Ineichen P, Seals R, Michalsky J, Stewart R. Modeling daylight availability and irradiance components from direct and global irradiance. Sol Energy 1990;44:271–89. https://doi.org/10.1016/0038-092X(90)90055-H.
- [48] Noorian AM, Moradi I, Kamali GA. Evaluation of 12 models to estimate hourly diffuse irradiation on inclined surfaces. Renew Energy 2008;33:1406–12. https:// doi.org/10.1016/j.renene.2007.06.027.
- [49] Kahl A, Dujardin J, Lehning M. The bright side of PV production in snow-covered mountains. Proc Nat Acad Sci 2019;116:1162–7. https://doi.org/10.1073/pnas. 1720808116.
- [50] Dobos A. PVWatts Version 5 Manual, Technical Report NREL/TP-6A20-62641, 1158421. National Renewable Eenrgy Laboratory; 2014. https://doi.org/10.2172/ 1158421.
- [51] Breiman L. Random forests. Mach Learn 2001;45:5–32. https://doi.org/10.1023/ A:1010933404324.
- [52] Huang G-B, Zhu Q-Y, Siew C-K. Extreme learning machine: theory and applications. Neurocomputing 2006;70:489–501. https://doi.org/10.1016/j.neucom.2005.12. 126.
- [53] Wang H, He Q, Shang T, Zhuang F, Shi Z. Extreme learning machine ensemble classifier for large-scale data. Cao J, Mao K, Cambria E, Man Z, Toh K-A, editors. Proceedings of ELM-2014. Proceedings in adaptation, learning and optimization, vol. 1. Cham: Springer International Publishing; 2015. p. 151–61. https://doi.org/ 10.1007/978-3-319-14063-6_14.
- [54] Breiman L. Bagging predictors. Mach Learn 1996;24:123–40. https://doi.org/10. 1007/BF00058655.

- [55] Heskes T. Practical confidence and prediction intervals. Advances in neural information processing systems, vol. 9. MIT Press; 1997. p. 176–82.
- [56] de Freitas Viscondi G, Alves-Souza SN. A Systematic Literature Review on big data for solar photovoltaic electricity generation forecasting. Sustain Energy Technol Assess 2019;31:54–63. https://doi.org/10.1016/j.seta.2018.11.008.
- [57] Rehman S, Ghori SG. Spatial estimation of global solar radiation using geostatistics. Renew Energy 2000;21:583–605. https://doi.org/10.1016/S0960-1481(00) 00078-1.
- [58] Pedregosa F, Varoquaux G, Gramfort A, Michel V, Thirion B, Grisel O, et al. Scikitlearn: Machine Learning in Python. J Mach Learn Res 2011;12:2825–30.
- [59] Zaksek K, Ostir K, Kokalj Z. Šky-view factor as a relief visualization technique. Remote Sens 2011;3:398–415. https://doi.org/10.3390/rs3020398.
- [60] Wan C, Xu Z, Pinson P, Dong ZY, Wong KP. Probabilistic forecasting of wind power generation using extreme learning machine. IEEE Trans Power Syst 2014;29:1033–44. https://doi.org/10.1109/TPWRS.2013.2287871.
- [61] Walch A, Castello R, Mohajeri N, Guignard F, Kanevski M, Scartezzini J-L. Spatiotemporal modelling and uncertainty estimation of hourly global solar irradiance using Extreme Learning Machines. Energy Procedia 2019;158:6378–83. https://doi. org/10.1016/j.egypro.2019.01.219.
- [62] JCGM. Evaluation of measurement data Guide to the expression of uncertainty in measurement. Technical Report JCGM 100:2008. Joint Committee for Guides in Metrology; 2008.
- [63] Goodman LA. The variance of the product of K random variables. J Am Stat Assoc 1962;57:54–60. https://doi.org/10.2307/2282440.
- [64] Portmann M, Galvagno-Erny D, Lorenz P, Schacher D. Sonnendach.ch: Berechnung von Potenzialen in Gemeinden, Technical Report, Bundesamt f
 ür Energie BFE; 2016.
- [65] SwissGrid, Production and Consumption; 2019. https://www.swissgrid.ch/en/ home/operation/grid-data/generation.html.

- [66] Swiss Federal Institue for Energy. Schweizerische Elektrizitätsstatistik 2018. Technical Report. Bundesamt für Energie BFE; 2018.
- [67] Walch A, Mohajeri N, Scartezzini J-L. A critical comparison of methods to estimate solar rooftop photovoltaic potential in Switzerland. J Phys: Conf Ser 2019;1343:012035. https://doi.org/10.1088/1742-6596/1343/1/012035.
- [68] Castello R, Roquette S, Esguerra M, Guerra A, Scartezzini J-L. Deep learning in the built environment: automatic detection of rooftop solar panels using Convolutional Neural Networks. J Phys: Conf Ser 2019;1343:012034. https://doi.org/10.1088/ 1742-6596/1343/1/012034.
- [69] Gulin M, Vašak M, Baotic M. Estimation of the global solar irradiance on tilted surfaces. In: 17th international conference on electrical drives and power electronics (EDPE 2013), vol. 6; 2013. p. 347–53.
- [70] Duffie J, Beckman W. Solar engineering of thermal processes: fourth edition, solar engineering of thermal processes: fourth edition. John Wiley and Sons; 2013. https://doi.org/10.1002/9781118671603.
- [71] Faiman D. Assessing the outdoor operating temperature of photovoltaic modules. Prog Photovoltaics Res Appl 2008;16:307–15. https://doi.org/10.1002/pip.813.
- [72] Holmgren WF, Hansen CW, Mikofski MA. pvlib python: a python package for modeling solar energy systems. J Open Source Softw 2018;3. https://doi.org/10. 21105/joss.00884.
- [73] Akusok A, Bjork K-M, Miche Y, Lendasse A. High-performance extreme learning machines: a complete toolbox for big data applications. IEEE Access 2015;3:1011–25. https://doi.org/10.1109/ACCESS.2015.2450498.
- [74] Jordahl K, den Bossche JV, Wasserman J, McBride J, Gerard J, Tratner J, et al. geopandas/geopandas: v0.5.0, 2019. https://doi.org/10.5281/zenodo.2705946. https://zenodo.org/record/2705946.
- [75] Neteler M, Bowman MH, Landa M, Metz M. GRASS GIS: a multi-purpose open source GIS. Environ Model Softw 2012;31:124–30. https://doi.org/10.1016/j. envsoft.2011.11.014.